Author's response to reviews

Title: Health-care services utilization and costs associated with radical cystectomy for bladder cancer: a descriptive population-based study in the province of Quebec, Canada

Authors:

- Fabiano Santos (fabiano.santos@mcgill.ca)
- Alice Dragomir (alice.dragomir@muhc.mcgill.ca)
- Ahmed S Zakaria (aszakaria81@yahoo.com)
- Wassim Kassouf (wassim.kassouf@muhc.mcgill.ca)
- Armen Aprikian (armen.aprikian@muhc.mcgill.ca)

Version: 2
Date: 12 January 2015

Author's response to reviews:

Manuscript was reviewed. Page and Line number was included.